Author's response to reviews

Title: Painful vertical diplopia as a presentation of a pituitary mass

Authors:

Shveta Bansal (shveta.ophth@gmail.com)
Kaveri Mandal (kmandal@doctors.org.uk)
Ahmed Kamal (ahmed.kamal@aintree.nhs.uk)

Version: 3 Date: 9 March 2007

Author's response to reviews:

Miss Shveta Bansal
MA (Hons) Cantab. MBBS. MRCS Ed
Senior House Officer, Royal Liverpool University Hospital Trust, UK
Email: shveta.ophth_@gmail.com
Tel: 0044- 0161 790 7969

9th march 2007

Dear Editor

Please find enclosed the re-submitted article as per requested by yourselves.

The following changes have been made:

1. Removed from title page: The text 'Title page', The manuscript type, The text 'Authors', The keywords section, job titles and qualifications.

2. Author list changed as requested.

3. Removed repeated title from above the abstract.

4. authors contributions included

5. Renamed case report 'case presentation'

6. Competing interest statement after conclusions

7. Figure labels changed from 1, 2a and 2b to 1, 2 and 3 and ammended in the manuscript.

Thanking you
Shveta Bansal
(Corresponding author)